

Author's response to reviews

Title: Addressing Inaccuracies on Emergency Medicine Residency Applications

Authors:

Eric D Katz (KATZE@MSNOTES.WUSTL.EDU)
Lee Shockley (lee.shockley@dhha.org)
Lawrence Kass (lkass@hmc.psu.edu)
David Howes (dhowes@medicine.bsd.uchicago.edu)
Janis P Tupesis (Janis_tupesis@hotmail.com)
Christopher Weaver (chsweave@iupui.edu)
Osman R Sayan (ors1@columbia.edu)
Jason Begue (jbegue@uabmc.edu)
Diamond Vrocher (dvrocher@uabmc.edu)
Jackie Frazer (jfrazer@uabmc.edu)
Timothy Evans (tcevans@hsc.vcu.edu)
H GENE Hern (gene_hern@yahoo.com)
Ralph Riviello (Ralph.riviello@jefferson.edu)
Antonio Rivera (riveraa@msnotes.wustl.edu)
Keith Kinoshita (kinoshik@msnotes.wustl.edu)
Edward Ferguson (fergused@msnotes.wustl.edu)

Version: 4 **Date:** 5 August 2005

Author's response to reviews:

To whom it may concern,

The name of one of the authors was netered incorrectly. I have corrected thi son the author list but did not correct it on the title page of the article. Antonio Rivera was mistakenly entered as Anthony Rivera. That is the only change to the manuscript since the revision of 8.4.5. I apologize for the error and inconvenience.

Eric D. Katz, MD